Reviewer 1:
A nicely written review article on Same day discharge following percutaneous cardiac interventions. The different procedures have been considered separately with evidence provided.

Answer to reviewer 1:
Thank you for reviewing our mini-review.

Reviewer 2:
The manuscript submitted by authors is interesting. However the showcasing of the data needs clarity in the manuscript. Following are the suggestions for the authors to improvise the manuscript. 1. The abstract should discuss or highlight the critical outcomes of the study. 2. 5 keywords to be suggested. Do not write short forms in the keywords and headings 3. The authors need to specify the search strategy adopted and timeline of the literature review carried out. The data included of the articles considered years from and to should be mentioned. 4. The authors should mention a section on future scope and perspectives 5. The authors can include the details of the previous studies in tabular form which will help readers to easily identify the outcomes. 6. Kindly ensure if full form of the abbreviated words are atleast presented once for first time in the manuscript.

Answer to reviewer 2:
1. The abstract delineates the trajectory of the article and states that individual procedures are reviewed in their respective sections. Do word limitations and complexity of the results, of each procedure are in their respective sections.
2. 5 keywords have been edited.
3. Search strategy section added to manuscript.
4. Section about future scopes added prior to conclusion.
5. Table 2 has been added which describes a summary of outcomes about same day discharge structural procedures included in the article.
6. All abbreviations have been identified and mentioned at first occurrence.